Author's response to reviews

Title: The association between adolescents' health and disparities in school career: a longitudinal cohort study

Authors:

Ellen Uiters (ellen.uiters@rivm.nl)
Erica Maurits (e.maurits@nivel.nl)
Mariël Droomers (m.droomers@amc.uva.nl)
Marieke Zwaanswijk (m.zwaanswijk@nivel.nl)
Robert A Verheij (r.verheij@nivel.nl)
Fons Van der Lucht (fons.van.der.lucht@rivm.nl)

Version: 3 Date: 24 April 2014

Author's response to reviews: see over
24 April 2014

Dear Mrs. Pafitis,

Please find enclosed a revised version of the manuscript ‘The association between adolescents’ health and disparities in school career: a longitudinal cohort study’, which I would like to submit for publication in BMC Public Health.

In line with your request, we have added a statement about the medical ethical approval of the study to page 6 (lines 136-139).

All authors have read and agreed to the content of the manuscript and have agreed to the submission. The manuscript contains original work that is not under consideration by any other journal. It has not already been published in any journal or other citable form.

Correspondence regarding the manuscript may be addressed to M. Zwaanswijk, NIVEL, P.O. Box 1568, 3500 BN Utrecht, The Netherlands; telephone: +31 30 2729791; fax: +31 30 2729729; e-mail: M.Zwaanswijk@nivel.nl.

In anticipation of your reaction, with kind regards,

Marieke Zwaanswijk, PhD
NIVEL, Netherlands Institute for Health Services Research
P.O. Box 1568
3500 BN Utrecht
The Netherlands